# Monoclonal antibody BTT1023 targeting vascular adhesion protein 1 for treating primary sclerosing cholangitis: BUTEO single-arm Phase II trial

Katherine Arndtz,¹ Yung-Yi Chen,¹ Anna Rowe,² Victoria Homer,² Amanda Kirkham,² Jessica Douglas-Pugh,² Daniel Slade,² Douglas Thorburn,³ Eleanor Barnes,⁴ Guruprasad Aithal,⁵ Philip Newsome,¹ David Smith,⁶ David Adams,¹ Christopher Weston¹ and Gideon Hirschfield¹\*

Declared competing interests of authors: The industry partner Acorda Therapeutics, Inc. (Ardsley, NY, USA) (formerly known as Biotie Therapies Ltd, Helsinki, Finland), funded the study drug and magnetic resonance imaging. David Smith is a company representative of Acorda Therapeutics, Inc., and was involved in the preclinical and clinical development of the investigational medicinal product and is an expert on vascular adhesion protein 1 (VAP-1) function. David Smith reports personal fees from Biotie Therapies Ltd (now part of Acorda Therapeutics, Inc.) outside the submitted work, Christopher Weston reports grants from the Medical Research Council (London, UK), the National Institute for Health Research Efficacy and Mechanism Evaluation programme and Biotie Therapies Ltd during the conduct of the study, and grants from C.H. Boehringer Sohn AG & Co. KG (Ingelheim am Rhein, Germany), Novo Nordisk A/S (Bagsværd, Denmark) and Pharmaxis Ltd (Frenchs Forest, NSW, Australia) outside the submitted work. In addition, Christopher Weston has a patent, Use of VAP-1 inhibitors for treating fibrotic conditions (WO2011029996A1), issued. Gideon Hirschfield reports consultancy from Acorda Therapeutics, Inc., CymaBay Therapeutics (Newark, CA, USA), GlaxoSmithKline plc (Brentford, UK), GENFIT (Loos, France), Intercept Pharmaceuticals, Inc. (New York, NY, USA), F. Hoffmann-La Roche Ltd (Basel, Switzerland), Pliant Therapeutics (South San Francisco, CA, USA) and Mirum Pharmaceuticals (Foster City, CA, USA) outside the submitted work.

<sup>&</sup>lt;sup>1</sup>Institute of Biomedical Research, NIHR Birmingham Biomedical Research Centre, University of Birmingham, Birmingham, UK

<sup>&</sup>lt;sup>2</sup>Cancer Research UK Clinical Trials Unit, University of Birmingham, Birmingham, UK <sup>3</sup>Liver Services, Royal Free London NHS Foundation Trust, London, UK

<sup>&</sup>lt;sup>4</sup>Nuffield Department of Medicine, University of Oxford, Oxford, UK

<sup>&</sup>lt;sup>5</sup>Faculty of Medicine and Health Sciences, The University of Nottingham, Nottingham, UK

<sup>&</sup>lt;sup>6</sup>Acorda Therapeutics, Inc., Ardsley, NY, USA

<sup>\*</sup>Corresponding author gideonhirschfield@gmail.com

Published March 2022 DOI: 10.3310/ZPNF4670

## **Plain English summary**

BUTEO single-arm phase II trial

Efficacy and Mechanism Evaluation 2022; Vol. 9: No. 1

DOI: 10.3310/ZPNF4670

NIHR Journals Library www.journalslibrary.nihr.ac.uk

## **Plain English summary**

Primary sclerosing cholangitis is an inflammatory and progressive liver disease. At present, there are no approved treatments for patients. The molecule vascular adhesion protein 1 has previously been shown in laboratory studies to be potentially involved in disease, including, in particular, in the development of liver scarring.

We wanted to test the idea that blocking this molecule with a drug given as an infusion would change markers of scarring in patient blood samples. To do so, we gave the drug BTT1023 (timolumab) by infusion to patients with primary sclerosing cholangitis who had elevated activity of the blood marker alkaline phosphatase. We gave the drug to 22 patients and carried out at a variety of blood tests before the drug was given and after seven infusions. We followed patients for any side effects of treatment. After 18 patients had been treated, we performed an interim analysis, which showed that, although the drug appeared safe, it was not possible to show any effect of infusions on liver inflammation. As a result, the study was subsequently ended.

Final analysis of all available tests has not shown an effect of this drug on relevant blood markers of inflammation or scarring in patients with primary sclerosing cholangitis.

In conclusion, although vascular adhesion protein 1 may play a role in the development of primary sclerosing cholangitis get disease, and how the disease progresses, blocking the activity of vascular adhesion protein 1 by treating patients with the drug BTT1023 had no effect on inflammation.

### **Efficacy and Mechanism Evaluation**

ISSN 2050-4365 (Print)

ISSN 2050-4373 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full EME archive is freely available to view online at www.journalslibrary.nihr.ac.uk/eme. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Efficacy and Mechanism Evaluation journal

Reports are published in *Efficacy and Mechanism Evaluation* (EME) if (1) they have resulted from work for the EME programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

#### **EME** programme

The Efficacy and Mechanism Evaluation (EME) programme funds ambitious studies evaluating interventions that have the potential to make a step-change in the promotion of health, treatment of disease and improvement of rehabilitation or long-term care. Within these studies, EME supports research to improve the understanding of the mechanisms of both diseases and treatments.

The programme supports translational research into a wide range of new or repurposed interventions. These may include diagnostic or prognostic tests and decision-making tools, therapeutics or psychological treatments, medical devices, and public health initiatives delivered in the NHS.

The EME programme supports clinical trials and studies with other robust designs, which test the efficacy of interventions, and which may use clinical or well-validated surrogate outcomes. It only supports studies in man and where there is adequate proof of concept. The programme encourages hypothesis-driven mechanistic studies, integrated within the efficacy study, that explore the mechanisms of action of the intervention or the disease, the cause of differing responses, or improve the understanding of adverse effects. It funds similar mechanistic studies linked to studies funded by any NIHR programme.

The EME programme is funded by the Medical Research Council (MRC) and the National Institute for Health Research (NIHR), with contributions from the Chief Scientist Office (CSO) in Scotland and National Institute for Social Care and Health Research (NISCHR) in Wales and the Health and Social Care Research and Development (HSC R&D), Public Health Agency in Northern Ireland.

#### This report

The research reported in this issue of the journal was funded by the EME programme as project number 12/165/31. The contractual start date was in January 2015. The final report began editorial review in September 2020 and was accepted for publication in July 2021. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The EME editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research. The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the MRC, NETSCC, the EME programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the EME programme or the Department of Health and Social Care.

Copyright © 2022 Arndtz et al. This work was produced by Arndtz et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

#### NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

#### **NIHR Journals Library Editors**

**Professor John Powell** Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editor-in-Chief of HSDR, PGfAR, PHR journals

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Adviser, Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

**Professor Helen Roberts** Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk